Title: Molecular prevalence of intestinal parasites infections in children with diarrhea in Franceville, southeast of Gabon.

Authors:

Sandrine Lydie Oyegue-Liabagui (lyds_ass@yahoo.fr)
Nal Kennedy Ndjangangoye (nndjangangoye@gmail.com)
Lady Charlene Kouna (charleneklc@gmail.com)
Gwaldys Mirlande Lekolo (edmirlande2000@yahoo.fr)
Franck Mounioko (fmounioko@yahoo.fr)
Sylvie Kwedi Nolna (bellasylk@yahoo.com)
Jean-Bernard Lekana-Douki (lekana jb@yahoo.fr)

Version: 5 Date: 23 Apr 2020

Author’s response to reviews:

Franceville, April 23, 2020

BMC Infectious Diseases
To the Editor-in-Chief

Dear Editor,

We would like to submit the revised manuscript entitled: “Molecular prevalence of intestinal parasites infections in children with diarrhea in Franceville, southeast of Gabon”, by OYEGUE-LIABAGUI SANDRINE LYDIE et al., which we would like to present for an original research. The manuscript number for the fourth revision was ID: INFD-D-19-02707R4.

Indeed, the manuscript had been submitted for the first time on December 23, 2019. However, this submission had not been validated due to the unavailability of the corresponding author. Following this situation, the manuscript had been submitted a second time on December 27, 2019 after change of the corresponding author. This problem of double submission had been resolved with Ms. Marielette. Thus the first submission of December 23, 2019 (with the correspondiing author Pr. Jean-Bernard Lekana-Douki) was deleted in favor of the second submission of December 27, 2019 (with the corresponding author Dr. Sandrine Lydie OYEGUE-LIABAGUI). Thus the corresponding author has been modified in your server as well as on the
Manuscript Number: INFD-D-19-02707
Full Title: Molecular prevalence of intestinal parasites infections in children with diarrhea in Franceville, southeast of Gabon
Corresponding Author: Sandrine Lydie OYEGUE-LIABAGUI
Universite des Sciences et Techniques de Masuku Franceville, Haut-Ogooué GABON
Corresponding Author E-Mail: lyds_ass@yahoo.fr

Manuscript Number: INFD-D-19-02707R1
Corresponding Author: Sandrine Lydie OYEGUE-LIABAGUI
Universite des Sciences et Techniques de Masuku Franceville, Haut-Ogooué GABON
Corresponding Author E-Mail: lyds_ass@yahoo.fr

Manuscript Number: INFD-D-19-02707R2
Corresponding Author: Sandrine Lydie OYEGUE-LIABAGUI
Universite des Sciences et Techniques de Masuku Franceville, Haut-Ogooué GABON
Corresponding Author E-Mail: lyds_ass@yahoo.fr

Manuscript Number: INFD-D-19-02707R3
Corresponding Author: Sandrine Lydie OYEGUE-LIABAGUI
Universite des Sciences et Techniques de Masuku Franceville, Haut-Ogooué GABON
Corresponding Author E-Mail: lyds_ass@yahoo.fr

Manuscript Number: INFD-D-19-02707R4
Corresponding Author: Sandrine Lydie OYEGUE-LIABAGUI
Universite des Sciences et Techniques de Masuku Franceville, Haut-Ogooué GABON
Corresponding Author E-Mail: lyds_ass@yahoo.fr

Thank you in advance for taking the time to review this revised manuscript.

Sincerely yours,

Sandrine Lydie OYEGUE-LIABAGUI, corresponding author
lyds_ass@yahoo.fr